Dear reviewers,

I will answer your questions one by one:

1. Rhabdomyolysis is a syndrome characterized by muscle necrosis and the release of muscle cell contents into the circulation. In mild cases, patients may not have elevated serum myokinase, but as the disease progresses, there may be a significant increase in creatine kinase, electrolyte disturbance, and acute kidney injury (AKI), which can be life-threatening.

2. Unfortunately, we considered that the weakness of the patient's limbs was caused by a metabolic disorder, and did not conduct an upper limb MRI examination.

3. I have deleted the tables and redundant language expressions, and re-polished the language. I am deeply sorry for the problems of my language expression.

Sincerely,

Ya-Han Wang